Author's response to reviews

Title: The use of complementary and alternative medicine after the completion of hospital treatment for colorectal cancer: Findings from a questionnaire study in Denmark

Authors:

Nina Nissen (ninissen@health.sdu.dk)
Anita Lunde (Anil@VIA.DK)
Christina G Pedersen (christina@psy.au.dk)
Helle Johannessen (hjohannessen@health.sdu.dk)

Version: 3 Date: 25 August 2014

Author's response to reviews: see over
Dear reviewers,

Many thanks for your additional comments to our manuscript. Below is the itemized response of how these comments have been addressed.

We trust that the manuscript will now be acceptable for publication.

**Reviewer 3:**

1. Abstract
   a) In our opinion, the verbs ‘to use’ and ‘to experience’ have different connotations, and we therefore have retained the sentence unchanged.
   b) Sentence ‘Of those using CAM ….’ has been amended.

2) Discussion
   a) The reference has been corrected
   b) Sentence ‘Beyond prevalence ….’ has been amended.

**Reviewer 4:**

As suggested, the aims of the study have been spelled out in the last paragraph of the introductory section.